Author’s response to reviews

Title: Adjuvant chemoradiation in pancreatic cancer: impact of radiotherapy dose on survival

Authors:
Alessio MORGANTI (amorganti60@gmail.com)
Francesco CELLINI (f.cellinimd@gmail.com)
Milly BUWENGE (mbuwenge@gmail.com)
Alessandra ARCELLI (alearceese@hotmail.it)
Sergio ALFIERI (sergio.alfieri@unicatt.it)
Felipe CALVO (felipe.calvo@salud.madrid.org)
Riccardo CASADEI (riccardo.casadei@unibo.it)
Savino CILLA (savinocilla@gmail.com)
Francesco DEODATO (francesco.deodato@fgps.it)
Giancarmine DI GIOIA (giancarmine.digioia@gmail.com)
Mariacristina DI MARCO (mariacristina.dimarco@unibo.it)
Lorenzo FUCCIO (lorenzofuccio@gmail.com)
Federica BERTINI (federica.bertini.rt@gmail.com)
Alessandra GUIDO (alessandraguido2008@gmail.com)
Joseph HERMAN (jherma15@jhmi.edu)
Gabriella MACCHIA (macchiagabriella@gmail.com)
Bert MAIDMENT (BWM5J@hsemail.mcc.virginia.edu)
Robert MILLER (miller.robert@mayo.edu)
Francesco MINNI (francesco.minni@unibo.it)
Paolo PASSONI (passoni.paolo@hsr.it)
Chiara VALENTINI (chiara.valentini83@gmail.com)
Author’s response to reviews:

RESPONSE TO THE EDITOR

Title: Adjuvant chemoradiation in pancreatic cancer: impact of radiotherapy dose on survival Journal: BMC Cancer

Submission ID: BCAN-D-19-00453

Comment 1:
Ethics approval and consent to participate
Thank you for your revision. Please ensure you have included the statement of consent to participate within this subsection.
Answer 1:
Thank you so much for your suggestion. We have included in the ethics approval and consent to participate subsection of the declarations section, lines 17, page 18 the following statement “All authors gave consent to participate. “

Comment 2:
Consent for publication
As your manuscript does not contain any individual persons’ data, please state “Not applicable” in this subsection
Answer 2:
Thank you much for your suggestion. In consent to participate of the declarations section, line 20, page 18, we have changed as suggested the following statement “Not applicable. “